ORIGINAL RESEARCH



Five-year Maintenance of Clinical Response and Consistent Safety Profile for Guselkumab in Asian patients with Psoriasis from VOYAGE 1 and VOYAGE 2

Byung Soo Kim · Seong Jin Jo · SangWoong Youn · Kristian Reich ·

Carine Saadoun \cdot Chia-Ling Chang \cdot Ya-Wen Yang \cdot Yu-Huei Huang \cdot

Tsen-Fang Tsai 🝺

Received: December 8, 2022 / Accepted: February 23, 2023 © The Author(s) 2023

ABSTRACT

Introduction: Guselkumab is a human monoclonal antibody against IL-23 used in the treatment of moderate-to-severe psoriasis. This posthoc analysis evaluated the efficacy and safety of guselkumab in the Asian subpopulation of VOYAGE 1 and VOYAGE 2 through 5 years. *Methods*: The proportions of guselkumab-treated Asian patients (VOYAGE 1 and 2) achieving Psoriasis Area and Severity Index (PASI) 90 and

Prior Publication: These data have not been published previously. However, the global data that include the Asian patient subgroup have been published: Baluvelt A et al. JAAD 2017;76(3):405–17; Reich K et al. JAAD 2016;76(3):418–31; Reick K et al. JAAD 2020;82(4):936–45; Reich K et al. Br J Dermatol 2021;185(6):1146–59.

B. S. Kim

Department of Dermatology, College of Medicine, Pusan National University, 179, Gudeok-ro, Seo-gu, Busan 49241, Republic of Korea

S. J. Jo

Department of Dermatology, Seoul National University Hospital, Seoul National University College of Medicine, 101, Daehak-ro, Jongno-gu, Seoul 03080, Republic of Korea

S. Youn

Department of Dermatology, Seoul National University Bundang Hospital, Seoul National University College of Medicine, 82 Gumi-ro 173Beon-gil, Bundang-gu, Seongnam-si, Gyeonggi-do 13620, Republic of Korea PASI 100, Investigator's Global Assessment (IGA) scores of 0/1 and 0, and Dermatology Life Quality Index (DLQI) scores of 0/1 (week 100 through week 252) were assessed. Non-responders were patients who met the treatment failure rules. Efficacy endpoints were analyzed using the as-observed methodology (no missing data imputation) for both studies and using non-responder imputation (for patients with any missing data) in VOYAGE 1. Safety outcomes were based on pooled data through week 252.

Results: Response rates through week 252 for 199 Asian patients in the guselkumab group in VOYAGE 1 and VOYAGE 2, respectively, were 76.8% and 80.6% (PASI 90), 26.8% and 38.7%

K. Reich

Translational Research in Inflammatory Skin Diseases, Institute for Health Services Research in Dermatology and Nursing, University Medical Center Hamburg-Eppendorf, Hamburg, Germany

C. Saadoun \cdot C.-L. Chang

Regional Medical Affairs Janssen Asia Pacific, Janssen, a division of Johnson & Johnson Pte Ltd., 2 Science Park Drive, #07-13, Ascent, Singapore Science Park 1, Singapore 118222, Singapore

Y.-W. Yang

Global Medical Affairs, Janssen Pharmaceutical Companies of Johnson & Johnson, LLC, Horsham, PA, USA (PASI 100), 64.3% and 87.1% (IGA 0/1), and 26.8% and 45.2% (IGA 0). DLQI (0/1) at week 252 was achieved by 52.7% of patients in VOYAGE 1 and 61.3% in VOYAGE 2, while DLQI (0) at week 252 was achieved by 32.7% of patients in VOYAGE 1 and 40.3% in VOYAGE 2. The safety profile was similar to the global population and remained consistent through 5 years. Asian patients were followed for a total of 814 patient-years (PY). Over 85% of the guselkumab-treated patients continued treatment through week 264. The rate of serious adverse events (AEs) at week 252 was 3.07/100 PY. Rates of AEs of interest were low: serious infections, 0.74/100 PY; nonmelanoma skin cancer (NMSC), no patients; malignancies other than NMSC, 0.12/100 PY; and no major adverse cardiovascular events (MACE).

Conclusion: These analyses confirm a continuous response over 5 years, indicating that guselkumab shows therapeutic longevity in Asian patients requiring long-term treatment for moderate-to-severe psoriasis.

Trial Registration: ClinicalTrials.gov identifiers: VOYAGE 1 [NCT02207231] and VOYAGE 2 [NCT02207244].

Y.-H. Huang

T.-F. Tsai (⊠) Department of Dermatology, National Taiwan University Hospital, No. 7, Chung-Shan S. Road, Taipei 100, Taiwan e-mail: tftsai@yahoo.com

PLAIN LANGUAGE SUMMARY

Psoriasis—a long-term condition that causes a skin rash with scaly, itchy patches (plaques)-is becoming more prevalent in Asia. To control symptoms of moderate-to-severe psoriasis and achieve a strong improvement in the patient's quality of life, continuous treatment is usually needed. Guselkumab is a medicine that targets specific parts of the immune system to treat moderate-to-severe psoriasis. It is important to understand the long-term benefits of guselkumab in Asian patient populations. Our study analyzed the data from two randomized clinical trials (called VOYAGE 1 and VOYAGE 2) that studied people with moderate-to-severe plaque psoriasis. We examined results for the 199 people from Asia, including Korea and Taiwan, who took part in these studies. Overall, 162 of the 184 (86.6%) people from Asia treated with guselkumab incorporated into these studies continued the treatment for 5 years. Patients treated with guselkumab showed effective clinical responses (improvements measured by clinicians), including high skin clearance, meaning a large reduction in skin surface area affected by psoriasis. On guselkumab, patients also reported improvements in their skin-related health-related quality of life. These improvements and the efficacy of guselkumab were maintained over 5 years of follow-up. The safety results for guselkumab in the Asian subpopulation were similar to those for the global population, showing low rates of serious adverse effects, as expected from this type of medicine. Overall, our study found a favorable benefit-risk profile with continuous guselkumab treatment for 5 years in Asian people with moderate-to-severe psoriasis. This highlights that guselkumab treatment allows long-lasting control of this disease.

Keywords: Asian; Biologics; Guselkumab; Moderate-to-severe; Psoriasis

Department of Dermatology, Chang Gung Memorial Hospital and College of Medicine, Chang Gung University, Guishan District, Taoyuan City, Taiwan

Key Summary Points

The prevalence of psoriasis is increasing in Asia, and continuous treatment is needed to maintain an optimal response.

While there are short-term efficacy and safety data for guselkumab in Asian patients with psoriasis, long-term data in this population are lacking.

This post-hoc analysis utilized Asian population data from two phase III, randomized, double-blind, placebo- and adalimumab comparator-controlled studies of guselkumab.

Continuous treatment with guselkumab over 5 years showed a favorable benefit-risk profile in Asian patients with moderate-to-severe psoriasis.

These findings indicate an agent with a potential for therapeutic longevity in the Asian population.

INTRODUCTION

Psoriasis is a serious global problem, with reported country prevalences ranging from 0 to 11.43% [1] and East Asian countries reporting an increasing prevalence of this chronic condition [2, 3]. Studies of psoriasis prevalence in China, Japan, South Korea and Taiwan estimate rates ranging from 0.12 to 2.14% [1-6]. As a chronic disease, moderate-to-severe psoriasis usually requires continuous treatment to maintain a response. Thus, long-term efficacy and safety data are crucial to inform the treatment-decision-making process. Another consideration is that with an increasingly aging population, effective treatments with a low risk of side effects are paramount [7]. Furthermore, studies in Asian and Western populations have reported differences in the expression of genes and human leukocyte antigen (HLA) that play a role in the pathophysiology and maybe even the treatment response of psoriasis [8], highlighting the importance of assessing the safety and efficacy of therapeutic agents specifically in the Asian population [9-13].

The current therapeutic options for this systemic inflammatory disease include topicals, phototherapy, conventional systemic treatments (e.g., methotrexate, cyclosporine), as well as biologics and small molecules [14]. In particular, biologics have transformed treatment for moderate-to-severe psoriasis, [7, 15], as there are drugs available that target IL-23 (guselkumab, risankizumab, tildrakizumab) and IL-17 (secukinumab) and anti-tumor necrosis factor (TNF) therapies (adalimumab) [14, 15].

Guselkumab is a fully human immunoglobulin G1 lambda monoclonal antibody. It binds with high affinity and specificity to the IL-23 p19 subunit and inhibits IL-23-mediated intracellular and downstream signaling [16]. As a key regulator of multiple cell types, IL-23 plays a crucial role in the pathogenesis of psoriasis [17, 18]. It is essential for the survival and expansion of T-helper 17 cells, which are part of the cascade contributing to skin inflammation in psoriasis [17, 18].

In two 48-week (1-year) phase III global clinical trials (VOYAGE 1 [NCT02207231] and VOYAGE 2 [NCT02207244]) that included both Asian and non-Asian populations, guselkumab demonstrated superior efficacy to placebo and adalimumab, a commonly used TNF- α inhibitor, and was well tolerated in patients with moderate-to-severe plaque psoriasis [19, 20]. Long-term guselkumab data have demonstrated the maintenance of clinical response and a consistent safety profile to week 156 (3 years) in patients from VOYAGE 1 and VOYAGE 2 [21], to week 204 (4 years) in patients from VOYAGE 1 [22], and to week 252 (5 years) in patients from both studies [23].

While a post-hoc analysis comparing the Asian and non-Asian subpopulations of VOY-AGE 1 and VOYAGE 2 found comparable responses for overall efficacy and safety in the short term (up to week 24) for these two subpopulations [24], data that are specific to longterm outcomes in the Asian subpopulation are needed. Here, we report the efficacy results and pooled safety data from a post-hoc analysis of the Asian subpopulation across VOYAGE 1 and VOYAGE 2 through 5 years of treatment with guselkumab.

METHODS

Patients

VOYAGE 1 and VOYAGE 2 were multinational trials including centers in the USA, Canada, Poland, Czech Republic, Germany, Spain, Russia, Australia, South Korea and Taiwan (VOY-AGE 1 only). For the purpose of this post-hoc analysis, patients recruited at centers located in Taiwan and South Korea were defined as Asian. Patients of Asian descent that participated at centers located in non-Asian countries were considered non-Asian due to the potential influence of differences in culture, health policies and types of treatments between Asian and non-Asian countries. Patient inclusion and exclusion criteria were identical for both trials. Eligible patients were ≥ 18 years of age, had a diagnosis of moderate-to-severe plaque psoriasis (defined as a score of ≥ 3 in the Investigator's Global Assessment $[IGA]_{,} > 12$ in the Psoriasis Area and Severity Index [PASI], and > 10% body surface area [BSA] involvement) for at least 6 months and were candidates for phototherapy or systemic psoriasis treatments (detailed inclusion and exclusion criteria are described elsewhere) [19].

Study Design

Details of the study designs have been published elsewhere [19–23]. Both VOYAGE 1 and VOYAGE 2 were phase III, multicenter, randomized, double-blind, placebo- and adalimumab comparator-controlled studies of guselkumab in moderate-to-severe psoriasis. While both studies evaluated the efficacy of guselkumab through week 252, and safety through week 264, VOYAGE 1 included a crossover to adalimumab at week 52 and VOY-AGE 2 included a randomized withdrawal and retreatment period at weeks 28–76 [23]. Both studies were conducted in accordance with ethical principles that have their origin in the Declaration of Helsinki and that are consistent with Good Clinical Practices. The study protocols were approved by institutional review boards or independent ethics committees, including Sterling Institutional Review Board. All participants provided written informed consent.

Efficacy and Safety Parameters

In VOYAGE 1 and VOYAGE 2, disease severity was evaluated based on standard psoriasis measures, including the PASI score and IGA, with evaluations performed at regular visits through week 252. Skin-related health-related quality of life (HRQoL) was self-reported by patients using dermatology-specific assessments, including the Dermatology Life Quality Index (DLQI). Safety was monitored by adverse event (AE) reporting and laboratory investigations.

In this Asian subpopulation (patients recruited at centers located in Taiwan and South Korea), analyses were conducted on a 'combined guselkumab group' using pooled data from VOYAGE 1 and VOYAGE 2, including placebo patients who crossed over to guselkumab and patients randomized to adalimumab at week 0 who crossed over to guselkumab at or after week 52. Prior biologic experience at baseline was also considered for this combined guselkumab group.

Statistical Analysis

This post-hoc analysis presents efficacy data from week 100 through week 252 from VOY-AGE 1 and VOYAGE 2. Although all patients from the two studies received guselkumab from week 76, cross-study comparisons at week 100 and beyond were considered to be the most informative, as they recognized the possibility that some patients withdrawn from treatment in VOYAGE 2 may have restarted or initiated guselkumab as late as week 72. At week 100, all patients had received a continuous regimen of guselkumab for at least 28 weeks, allowing for meaningful evaluation of efficacy outcomes across both studies.

The proportions of patients achieving nearly complete clearance (PASI 90 and an IGA score of 0/1), complete clearance (PASI 100 and an IGA score of 0), and no impact of psoriasis on HRQoL (DLQI of 0/1) are summarized. Efficacy data are presented for the guselkumab group (including patients randomly assigned to receive placebo at baseline who then crossed over to guselkumab at week 16) and the adalimumab-to-guselkumab group (patients randomly assigned to receive adalimumab who then crossed over to receive guselkumab) for each study.

In the prespecified analysis, patients who met treatment failure rules (TFR)-defined as discontinuation due to lack of efficacy, worsening of psoriasis, or the use of a protocol-prohibited psoriasis treatment-were considered non-responders, with no other missing data imputation for efficacy endpoints. To assess the robustness of the prespecified analyses, efficacy endpoints were analyzed with the as-observed methodology (no missing data imputation) for both studies and non-responder imputation (NRI) (patients with any missing data were counted as non-responders) in VOYAGE 1 (wherein patients received continuous treatment through week 252). Because of the randomized withdrawal design, NRI analyses were not performed for VOYAGE 2.

Safety data are summarized for patients receiving at least one dose of guselkumab. Data were pooled across the two studies and calculated as incidence per 100 patient-years (PY) of follow-up through week 264. These analyses focused mainly on the safety of guselkumab in patients receiving continuous treatment for up to week 252 (i.e., the guselkumab group, including placebo crossovers). Reported AEs included serious AEs, malignancies (nonmelanoma skin cancer [NMSC] and malignanother than NMSCs), major adverse cies cardiovascular events (MACEs), and serious infections. For completeness, cumulative safety data are presented for the combined guselkumab group (including the guselkumab and adalimumab-to-guselkumab groups).

RESULTS

Baseline Demographics

A total of 199 Asian patients were included in this post-hoc analysis: 101 patients in VOYAGE 1 (placebo, n = 22; guselkumab, n = 43; adalimumab, n = 36) and 98 in VOYAGE 2 (placebo, n = 23; guselkumab, n = 51; adalimumab, n = 24). Baseline demographics and disease characteristics were generally comparable across the two studies (Table 1). VOYAGE 2 had a higher proportion of patients with an IGA score of 'severe', and VOYAGE 1 had a higher proportion of patients with self-reported psoriatic arthritis.

Patient Disposition

Around 80% of all patients (Asian and non-Asian) in VOYAGE 1 and VOYAGE 2 continued through week 252. Overall, 19.6% of patients (152 of 774) in VOYAGE 1 and 23.4% (222 of 949) in VOYAGE 2 discontinued the study agent. Rates of discontinuation due to lack of efficacy were low for the guselkumab groups (1.4% in VOYAGE 1 and 1.0% in VOYAGE 2). An AE of worsening of psoriasis led to discontinuation in 0.4% of patients in VOYAGE 2 [23].

Over 85% of all Asian patients in VOYAGE 1 and VOYAGE 2 continued through week 252.

Overall, 9.6% of Asian patients (9 of 94) in VOYAGE 1 and 13.3% of Asian patients (12 of 90) in VOYAGE 2 discontinued the study agent from baseline (for those originally randomized to guselkumab) and from the time of crossover to guselkumab (for those originally randomized to placebo or adalimumab) through week 252. Rates of discontinuation due to lack of efficacy were low for the guselkumab groups (1.6% in VOYAGE 1, and no patients discontinued due to lack of efficacy in VOYAGE 2). There were no patient discontinuations due to an AE of worsening of psoriasis in either VOYAGE 1 or VOY-AGE 2 for the Asian subpopulation.

	VOYAGE 1 $(n = 101)$		VOYAGE 2 $(n = 98)$			
	Placebo	Guselkumab	Adalimumab	Placebo	Guselkumab	Adalimumab
Patients randomized, <i>n</i>	22	43	36	23	51	24
Age (years), mean \pm SD	42.1 (13.33)	40.7 (11.85)	38.3 (11.84)	43.1 (10.28)	41.6 (12.59)	37.9 (7.10)
Men, n (%)	14 (63.6%)	33 (76.7%)	30 (83.3%)	17 (73.9%)	40 (78.4%)	20 (83.3%)
Weight (kg)						
Mean \pm SD	73.7 (14.94)	77.3 (17.88)	82.5 (16.31)	75.0 (14.02)	75.7 (17.19)	77.6 (14.51)
> 90 kg, n (%)	3 (13.6%)	8 (18.6%)	9 (25.0%)	3 (13.0%)	7 (13.7%)	4 (16.7%)
> 70 to ≤ 90 kg, <i>n</i> (%)	8 (36.4%)	18 (41.9%)	21 (58.3%)	11 (47.8%)	26 (51.0%)	13 (54.2%)
\leq 70 kg, <i>n</i> (%)	11 (50.0%)	17 (39.5%)	6 (16.7%)	9 (39.1%)	18 (35.3%)	7 (29.2%)
Height (cm), mean \pm SD	167.9 (7.09)	168.7 (9.09)	168.9 (9.09)	169.1 (8.77)	169.7 (8.02)	171.2 (8.08)
BMI (kg/m ²), mean \pm SD	26.1 (4.72)	27.1 (5.68)	28.8 (4.85)	26.1 (3.71)	26.1 (4.75)	26.4 (3.81)
Duration of psoriasis (years), mean \pm SD	12.6 (6.71)	15.2 (10.47)	13.2 (7.24)	12.6 (6.07)	15.3 (9.38)	10.4 (6.69)
Psoriatic arthritis, n (%)	5 (22.7%)	9 (20.9%)	7 (19.4%)	2 (8.7%)	3 (5.9%)	3 (12.5%)
BSA (%), mean \pm SD	32.7 (16.75)	35.6 (18.22)	34.8 (20.84)	32.6 (18.64)	30.5 (14.33)	33.5 (22.13)
PASI (0–72), mean \pm SD	24.94 (9.945)	26.16 (9.130)	27.72 (12.164)	24.24 (8.647)	23.45 (8.804)	25.40 (13.397)
IGA score						
Moderate, n (%)	16 (72.7%)	33 (76.7%)	24 (66.7%)	15 (65.2%)	35 (68.6%)	15 (62.5%)
Severe, n (%)	6 (27.3%)	10 (23.3%)	12 (33.3%)	8 (34.8%)	16 (31.4%)	9 (37.5%)
DLQI (0–30), mean \pm SD	18.1 (6.34)	19.0 (7.18)	18.5 (6.82)	19.4 (7.25)	17.3 (6.17)	18.7 (7.11)
Prior psoriasis treatment, n (%)						
Phototherapy (PUVA or UV-B)	14 (63.6%)	38 (88.4%)	30 (83.3%)	22 (95.7%)	44 (86.3%)	19 (79.2%)

Table 1 Summary of patient demographics, characteristics, and disease characteristics in Asian patients in VOYAGE 1 andVOYAGE 2

	VOYAGE	1 (n = 101)		VOYAGE 2 $(n = 98)$		
	Placebo	Guselkumab	Adalimumab	Placebo	Guselkumab	Adalimumab
Non-biologic systemics*	16 (72.7%)	41 (95.3%)	32 (88.9%)	19 (82.6%)	46 (90.2%)	22 (91.7%)
Biologics [†]	5 (22.7%)	9 (20.9%)	10 (27.8%)	7 (30.4%)	17 (33.3%)	6 (25.0%)
Non-biologic systemics or biologics	17 (77.3%)	41 (95.3%)	32 (88.9%)	20 (87.0%)	46 (90.2%)	23 (95.8%)

Table 1 continued

*Non-biologics include PUVA, methotrexate, cyclosporin, acitretin, apremilast, or tofacitinib. †Biologics include etanercept, infliximab, alefacept, efalizumab, ustekinumab, briakinumab, secukinumab, ixekizumab, or brodalumab

BMI body mass index, *BSA* body surface area, *DLQI* Dermatology Life Quality Index, *IGA* Investigator's Global Assessment, *PASI* Psoriasis Area and Severity Index, *PUVA* psoralen plus ultraviolet A, *SD* standard deviation, *UV-B* ultraviolet B

Clinical Response

Clinical responses in Asian patients treated with guselkumab were maintained over time in both VOYAGE 1 and VOYAGE 2, demonstrating consistent results from week 100 through week 252. In the prespecified TFR analysis of VOY-AGE 1, the proportion of patients achieving a PASI 90 response in the guselkumab group at weeks 100 and 252 was, respectively, 76.2% and 76.8%, and the proportion achieving a PASI 100 response at weeks 100 and 252 was, respectively, 34.9% and 26.8% (Fig. 1A). Similarly, in VOYAGE 2, the PASI 90 and PASI 100 responses at week 100 (72.1% and 26.5%, respectively) were maintained through week 252 (80.6% and 38.7%, respectively) (Fig. 1B). The PASI responses for patients treated with adalimumab who crossed over to guselkumab were also maintained over time in both VOYAGE 1 and VOY-AGE 2 and were generally comparable to the responses in the guselkumab group (Fig. 1A, B).

Similar responses were observed for IGA scores based on the prespecified TFR analysis. An IGA score of 0/1 at weeks 100 and 252 was achieved by 74.6% and 64.3% of patients, respectively, in VOYAGE 1, and by 80.9% and 87.1% of patients in VOYAGE 2 (Fig. 1C, D). The proportion of patients achieving an IGA score of 0 in the guselkumab group at weeks 100 and 252 was 41.3% and 26.8%, respectively, in VOYAGE 1, and 39.7% and 45.2% in VOYAGE 2

(Fig. 1C, D). IGA scores for patients treated with adalimumab who crossed over to guselkumab were also maintained over time in both VOY-AGE 1 and VOYAGE 2, and were generally comparable to responses in the guselkumab group.

Non-responder Imputation and As-Observed Analyses

PASI and IGA outcomes were also analyzed using NRI in VOYAGE 1 and the as-observed methodology through week 252 in VOYAGE 1 and VOYAGE 2. Compared to the prespecified TFR analysis, the proportions of patients achieving PASI 90, PASI 100, an IGA score of 0/1 and an IGA score of 0 at week 252 were similar for the as-observed analyses in both VOYAGE 1 (78.2%, 27.3%, 65.5%, 27.3%, respectively) and VOYAGE 2 (80.6%, 38.7%, 87.1%, 45.2%, respectively), and were slightly lower when the NRI rules were considered in VOYAGE 1 (67.2%, 23.4%, 56.3%, 23.4%, respectively); see Fig. 2A, B. In both analyses, the response rates remained generally stable over time. NRI and as-observed analyses for patients treated with adalimumab who crossed over to guselkumab also showed a similar pattern.



Fig. 1 Clinical response by treatment group through week 252 across VOYAGE 1 and VOYAGE 2 (prespecified TFR analysis). **A** PASI 90 and PASI 100 response in VOYAGE 1. **B** PASI 90 and PASI 100 response in VOYAGE 2. **C** IGA scores of 0/1 (cleared/minimal) and 0 (cleared) in VOYAGE 1. **D** IGA scores of 0/1 and 0 in VOYAGE 2. *In the prespecified analysis, patients who

Patient-Reported Outcomes

Patient-reported outcomes (DLQI score 0/1 and 0) substantiated the improvements in clinical responses to guselkumab (Table 2). In the guselkumab group, DLQI scores of 0/1 at weeks 100 and 252, respectively, were achieved by 53.2% and 52.7% of patients in VOYAGE 1, and by 47.1% and 61.3% of patients in VOYAGE 2. The proportion of patients achieving a DLQI of 0 in the guselkumab group at weeks 100 and 252, respectively, was 30.6% and 32.7% in VOYAGE 1 and 36.8% and 40.3% in VOYAGE 2. The mean (standard deviation) change from

met TFR (defined as discontinuation due to lack of efficacy, worsening of psoriasis, or the use of a protocolprohibited psoriasis treatment) were considered nonresponders. *ADA* adalimumab, *GUS* guselkumab, *IGA* Investigator's Global Assessment, *PASI* Psoriasis Area and Severity Index, *TFR* treatment failure rules

baseline DLQI score at weeks 100 and 252, respectively, was -15.0 (7.29) and -15.3 (7.49) in VOYAGE 1 and -14.8 (6.73) and -14.7 (6.93) in VOYAGE 2. DLQI scores for patients treated with adalimumab who crossed over to guselkumab were also maintained over time in both VOYAGE 1 and VOYAGE 2 and were generally comparable to responses in the guselkumab group. Of note, at week 252, a larger proportion of patients in the guselkumab group had a DLQI score of 0/1 than in the crossover group: 52.7%% vs 40.0% (VOYAGE 1), and 61.3% vs 37.5% (VOYAGE 2).



Fig. 1 continued

Safety Outcomes

Asian patients were followed for a total of 814 patient-years (PY). The AE rate per 100 PY was 148.08/100 PY (95% CI 139.84, 156.68) across studies, and the number of discontinuations due to AEs was 0.74/100 PY (95% CI 0.27, 1.60). The rate of serious AEs was 3.07/100 PY (95% CI 1.99, 4.53). Rates of AEs of interest were low: serious infections, 0.74/100 PY (95% CI 0.27, 1.60); none of the patients experienced NMSC; malignancies other than NMSC, 0.12/100 PY (95% CI: 0.00, 0.68); no MACE was reported (Table 3). Guselkumab exposure per 100 PY over time also showed a consistent safety profile

from year 1 through year 5 for all AEs of interest. The safety profile for guselkumab in the Asian subpopulation was similar to that of the global population: AE rate per 100 PY, 149.0/ 100 PY (95% CI: 147.0, 152.0), and rate of serious AEs, 5.01/100 PY (95% CI 4.50, 5.56) [25].

DISCUSSION

In the overall global population, over 75% of all patients from both studies continued treatment through year 5 [23]; for Asian patients treated with guselkumab, over 85% of patients continued treatment through 5 years. This analysis of the Asian subpopulation across both studies



◄ Fig. 2 Clinical response by analysis type in the gusekumab group through week 252 in VOYAGE 1. A PASI 90 and PASI 100. B IGA scores of 0/1 (cleared/minimal) and 0 (cleared). *In the prespecified TFR analyses, patients who met TFR (defined as discontinuation due to lack of efficacy, worsening of psoriasis, or the use of a protocolprohibited psoriasis treatment) were considered nonresponders.[†]In the non-responder imputation analyses, patients with missing efficacy data after the application of TFR were counted as non-responders regardless of the reason for the missing data.[‡]In the as-observed analyses, the data available at each visit were used and missing data was not imputed. *IGA* Investigator's Global Assessment, *PASI* Psoriasis Area and Severity Index, *TFR* treatment failure rules

shows maintenance dosing with guselkumab 100 mg every 8 weeks sustained high levels of clinical response in the majority of patients through 5 years.

Treatment effects for IGA and PASI responses at weeks 16 and 24 have previously been shown to be consistent between the Asian and non-Asian populations in VOYAGE 1 and VOYAGE 2 [24]. At week 16, treatment differences between guselkumab and placebo were significant for IGA 0/1 and PASI 90 responses (co-primary endpoints) in both the Asian and non-Asian populations and were comparable between populations [24]. At week 24, treatment with guselkumab was superior to adalimumab for IGA 0/1 and PASI 90, and these responses were comparable between the Asian and non-Asian populations [24].

Based on this current prespecified TFR analysis of VOYAGE 1, over 75% of Asian patients exhibited a durable PASI 90 response, and over 60% presented a durable IGA score of 0/1, with complete skin clearance maintained by over 25% of Asian patients through 5 years without dose escalation (evaluated by PASI 100 or IGA score of 0 assessments). In the prespecified TFR analysis of VOYAGE 2, durable PASI 90 and IGA score of 0/1 responses were exhibited by over 80% and almost 90% of Asian patients, respectively, and complete skin clearance was maintained by 45% (IGA score of 0) and 38% (PASI 100) of Asian patients through 5 years. Applying the more conservative NRI methodology yielded slightly lower response rates compared with both the TFR and as-observed analyses of continual treatment with guselkumab (VOY-AGE 1). However, high response rates were maintained through 5 years regardless of the analysis methodology. This attests to the robustness of these data and confirms the durability of guselkumab therapy in the Asian population.

In the global VOYAGE 1 and VOYAGE 2 populations, a PASI 90 response was achieved by 84.1% and 82.0% of guselkumab-treated patients at week 252, respectively; a PASI 100 response was achieved by 52.7% and 53.0% of patients; and an IGA score of 0/1 response was achieved by 82.4% and 85.0% of patients [22, 23]. Except for a numerically higher IGA 0/1 response in VOYAGE 2, the PASI and IGA responses in the Asian population were consistent, albeit pared down for PASI 100 with respect to the global population. This may have been due to the greater proportion of Asian patients with more severe psoriasis at baseline (a higher percentage of BSA involvement, a higher baseline PASI and a higher proportion of patients with a baseline IGA score of 4 [severe]), as well as the greater likelihood of Asian patients having undergone phototherapy and/ or systemic therapies compared with non-Asian patients [24].

Consistent with the clinical responses, improvements were maintained in patient-reported HRQoL as measured by DLQI. Previous reports of treatment effects for the DLQI score of 0/1 response in VOYAGE 1 and VOYAGE 2 between guselkumab and placebo treatments and between guselkumab and adalimumab treatments showed a difference between the Asian and non-Asian subpopulations at week 16 [24]. A smaller proportion of the Asian subpopulation achieved a DLQI score of 0/1 response compared to the non-Asian subpopulation, regardless of treatment [24]. A similar trend was observed at week 24, although the treatment differences between guselkumab and adalimumab were comparable between populations [24]. The authors suggested that this may have potentially been due to the higher baseline DLQI in all treatment groups in the Asian subpopulation than in the non-Asian

DLQI	Week 10	00	Week 15	56	Week 17	72	Week 20	4	Week 228		Week 25	5
outcomes	GUS*	ADA → GUS†	GUS*	$\substack{ADA\\ \rightarrow GUS^{\dagger}}$	GUS*	ADA → GUS†	GUS*	ADA → GUS†	GUS*	ADA → GUS†	GUS*	ADA → GUS†
VOYAGE 1												
Baseline score $> 1, n$	62	30	60	30	58	30	58	30	57	30	55	30
Score = $0/1$, n (%)	33 (53.2)	16 (53.3)	36 (60.0)	14 (46.7)	35 (60.3)	16 (53.3)	39 (67.2)	13 (43.3)	36 (63.2)	14 (46.7)	29 (52.7)	12 (40.0)
Score = 0, n (%) VOYAGE 2	19 (30.6)	10 (33.3)	22 (36.7)	8 (26.7)	24 (41.4)	10 (33.3)	24 (41.4)	9 (30.0)	27(47.4)	10 (33.3)	18 (32.7)	9 (30.0)
Baseline score > 1 , n	68	17	64	16	62	16	63	16	63	16	62	16
Score = $0/1$, n (%)	32 (47.1)	9 (52.9)	30 (46.9)	7 (43.8)	31 (50.0)	8 (50.0)	33 (52.4)	9 (56.3)	35 (55.6)	7 (43.8)	38 (61.3)	6 (37.5)
Score = 0, n (%)	25 (36.8)	6 (35.3)	22 (34.4)	5 (31.3)	23 (37.1)	6 (37.5)	21 (33.3)	6 (37.5)	23 (36.5)	7 (43.8)	25 (40.3)	6 (37.5)
*Placebo crossov *This group incl	er patients udes patien	were included ts randomly a	l in the gu ssigned to r	selkumab colu received adalin	imn after c numab at be	rossover to gu aseline who cr	ıselkumab rossed over 1	to receive gu	selkumab at	t or after weel	c 52 in VO	YAGE 1 and

at or after week 28 in VOYAGE 2 *ADA* adalimumab, *DLQI* Dermatology Life Quality Index, *GUS* guselkumab

	Guselkumab ^a (Asian subpopulation)	Adalimumab → guselkumab ^b (Asian subpopulation)	Combined guselkumab (Asian subpopulation)
Patients treated with guselkumab, <i>n</i>	136	48	184
Total patient-years of follow-up	623	192	814
Adverse events	156.77 (147.09, 166.92)	119.88 (104.89, 136.42)	148.08 (139.84, 156.68)
Adverse events leading to discontinuations	0.96 (0.35, 2.10)	0.00	0.74 (0.27, 1.60)
Infections	53.17 (47.59, 59.21)	41.18 (32.60, 51.32)	50.34 (45.59, 55.46)
Infections requiring treatment	19.11 (15.83, 22.87)	13.55 (8.85, 19.86)	17.80 (15.02, 20.95)
Serious adverse events	3.69 (2.34, 5.54)	1.04 (0.13, 3.77)	3.07 (1.99, 4.53)
Serious infections	0.80 (0.26, 1.87)	0.52 (0.01, 2.90)	0.74 (0.27, 1.60)
Malignancies	0.16 (0.00, 0.90)	0.00	0.12 (0.00, 0.68)
Nonmelanoma skin cancer	0.00	0.00	0.00
Other than nonmelanoma skin cancer	0.16 (0.00, 0.90)	0.00	0.12 (0.00, 0.68)
Major adverse cardiovascular events ^c	0.00	0.00	0.00

Table 3 Average incidence of adverse events in guselkumab-treated Asian patients per 100 patient-years through week 264by treatment group in VOYAGE 1 and VOYAGE 2 studies

Rates are reported as number of events per 100 patient-years (95% confidence interval) for all adverse events except malignancies, which are reported as number of patients with events per 100 patient-years (95% confidence interval)

^aIncludes patients randomized to guselkumab at baseline and those randomized to placebo at baseline who crossed over to receive guselkumab at week 16 in VOYAGE 1 and VOYAGE 2

^bIncludes patients randomized to adalimumab at baseline who crossed over to receive guselkumab either at week 52 in VOYAGE 1 or at or after week 28 in VOYAGE 2

^cMajor adverse cardiovascular events include cardiovascular death, nonfatal myocardial infarction events, and nonfatal stroke

subpopulation [24]. In the current analysis, the DLQI score of 0/1 response was maintained through 5 years by over 50% of Asian patients in VOYAGE 1 and over 60% of Asian patients in VOYAGE 2. The lower DLQI response in VOY-AGE 1 may have been due to a higher level of self-reported psoriatic arthritis at baseline. At year 5, a larger proportion of patients in the guselkumab group had a DLQI score of 0/1 than in the adalimumab crossover group. This is most likely a result of patients in the

guselkumab group receiving continuous guselkumab treatment for a longer duration than those in the adalimumab crossover group. Together with the clinical responses, these observations show maintenance of response of guselkumab from the perspectives of both the patient and physician.

In the global population, the safety profile remained consistent and favorable through 5 years of continuous guselkumab treatment. The AE and serious AE rates were 149/100 PY and 5.01/100 PY, respectively, with low rates of AEs of interest ($\leq 0.85/100$ PY), including NMSC, malignancies other than NMSC, and MACE [25]. Similar findings were evident for Asian patients, with rates of serious AEs and AEs of interest remaining low through 5 years. There were no reports of malignancies, MACE or deaths through 5 years.

In this study, patients of Asian descent participating at centers located in non-Asian countries were considered non-Asian for a number of reasons. Psoriasis may be associated with an even higher psychosocial burden in Asian countries than in non-Asian countries due to cultural reasons, socioeconomic status, and higher levels of social stigma resulting from misunderstanding and misconceptions [26]. Policy and decision makers in Asia consider psoriasis to be a moderate dermatological condition, placing it lower on healthcare priority lists than in non-Asian countries. The management of psoriasis in Asia is further complicated by the use of traditional and herbal medicines [26].

Both VOYAGE 1 and VOYAGE 2 had high retention rates for patients throughout the 5 years. Rates of discontinuation through week 252 were very low for patient populations in both VOYAGE 1 (global 19.6%; Asian 9.6%) and VOYAGE 2 (global 23.4%; Asian 13.3%), indicating that discontinuations were not a major issue in these studies [23]. Furthermore, a proactive approach was taken to address missing data in the prespecified analyses by assigning non-responder status to patients who discontinued the study agent due to lack of efficacy or worsening of psoriasis, and to those who initiated protocol-prohibited psoriatic medications.

Encouraging findings from two real-life retrospective studies conducted in Italy have confirmed the effectiveness and safety of guselkumab in daily clinical practice up to 3 years [27], and as a potential treatment option in patients who have previously failed on anti-IL-17 therapy [15]. How these real-life findings translate to Asian patients with moderate-tosevere psoriasis is worthy of future studies, as is the advancement of personalized medicine in tailoring biologics to these individuals with psoriasis [14].

Limitations of this post-hoc analysis include the lack of blinding and comparator arms during the extension period of both studies. While this may have led to reporting bias, the various analyses conducted minimize this potential limitation, and so these data provide a valuable insight into the ongoing effects of long-term guselkumab treatment for psoriasis in the Asian population.

CONCLUSION

These data through 5 years of continuous treatment with guselkumab in the VOYAGE 1 and VOYAGE 2 trials confirm the favorable benefit–risk profile of this agent in Asian patients with moderate-to-severe psoriasis. Continued responses over 5 years indicate therapeutic longevity for Asian patients requiring long-term treatment for this chronic condition.

ACKNOWLEDGEMENTS

The authors would like to thank the participants of the study.

Medical Writing and Editorial Assistance Medical writing support, in accordance with GPP guidelines, was provided by Mediwrite Asia Inc Pte Ltd, and included editorial and proofreading services funded by Janssen Asia Pacific.

Author Contributions. Byung Soo Kim, Seong Jin Jo, SangWoong Youn, Kristian Reich, Carine Saadoun, Chia-Ling Chang, Ya-Wen Yang, Yu-Huei Huang, and Tsen-Fang Tsai contributed to the concept, drafting and revisions of this manuscript, and all authors have read and approved the final manuscript.

Funding. This pooled analysis was funded by Janssen Asia Pacific, a division of Johnson and Johnson Pte Ltd. Janssen Asia Pacific also provided funding and guidance for the preparation of this article, including the analysis of the data, the writing of this report and for the Journal Rapid Service fee. The authors wish to thank Yuhan Huang, Senior Principal Statistician from Statistics & Decision Sciences, and Changchun Cheng, Senior Statistical Programmer Analyst from Clinical & Statistical Programming, for performing the analysis.

Data Availability. The data sharing policy of Janssen Pharmaceutical Companies of Johnson & Johnson is available at https://www. janssen.com/clinical-trials/transparency. As noted on that site, requests for access to the datasets generated during and/or analyzed during the current study can be submitted through the Yale Open Data Access (YODA) Project site at http://yoda.yale.edu.

Declarations

Conflicts of Interest. Byung Soo Kim has served as a scientific adviser, a clinical study investigator and/or a speaker for AbbVie, Astellas, Boehringer Ingelheim, Bristol Myers Squibb, Celltrion, Eli Lilly, Galderma, GlaxoSmithKline, Janssen, Kyowa Hakko Kirin, LEO Pharma, Novartis, Regeneron, Samsung, Sanofi and UCB. Seong-Jin Jo has conducted clinical trials for, acted as a consultant for, or received speaker's honoraria from AbbVie, Boehringer Ingelheim, Bristol Myers Squibb, Daewoong, Janssen, LEO Pharma, Lilly, Novartis, Pfizer and Sanofi. SangWoong Youn served as a speaker for Abb-Vie, Cellgene, Eli Lilly, Janssen, LEO Pharma, Novartis and Pfizer organised psoriasis symposium or meetings, and has also performed phase III clinical trials sponsored by Janssen, Novartis, Boehringer Ingelheim, BMS, Eli Lilly, Kyowa Kirin and UCB and phase IV clinical trials sponsored by CKD Pharma, Janssen and LEO Pharma. Kristian Reich has served as a paid advisor and/or a paid speaker for and/or participated in clinical trials (site received patient fees, received fees if acting as a coordinating investigator) sponsored by AbbVie, Affibody, Biogen-Idec, Almirall, Amgen, Boehringer Ingelheim Pharma, Celgene, Covagen, Forward Pharma, Fresenius Medical Care, Galderma, GlaxoSmithKline, Janssen, Janssen-Cilag,

Kyowa Kirin, Leo, Eli Lilly, Medac, Merck Sharp & Dohme, Miltenvi, Novartis, Ocean Pharma, Pfizer, Samsung Bioepis, Sanofi, Takeda, UCB Pharma, Valeant and Xenoport. Carine Saadoun, Chia-Ling Chang and Ya-Wen Yang are employees of Janssen Pharmaceutical Companies of Johnson & Johnson. Yu-Huei Huang has conducted clinical trials while serving as a principal investigator for AbbVie, Galderma, Eli Lilly, Janssen-Cilag Pharmaceutica, Novartis Pharmaceuticals Corporation and Pfizer and has received honoraria for serving as an advisory board member for and speaking fees from Abb-Vie, Celgene, Eli Lilly, Janssen-Cilag Pharmaceutica, Novartis Pharmaceuticals Corporation and Pfizer Limited. Tsen-Fang Tsai has conducted clinical trials or received honoraria for serving as a consultant for Abbvie, Boehringer Ingelheim, Bristol Myers Squibb, Celgene, Eli-Lilly, Galderma, GSK, Janssen-Cilag, Leo Pharma, Merck Sharp & Dohme, Novartis International, Pfizer, PharmaEssentia, Sanofi, and UCB Pharma.

Ethical Approval. This article is based on previously conducted studies (VOYAGE 1 and VOYAGE 2) and does not contain any new studies with human participants performed by any of the authors. Both VOYAGE 1 and VOY-AGE 2 were conducted in accordance with ethical principles that have their origin in the Declaration of Helsinki and are consistent with Good Clinical Practices. The study protocols were approved by institutional review boards or independent ethics committees. All participants provided written informed consent.

This article is licensed under **Open** Access. a Creative Commons Attribution-NonCommercial 4.0 International License, which persharing, mits anv non-commercial use, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If

material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http:// creativecommons.org/licenses/by-nc/4.0/.

REFERENCES

- Bihl M, Tamm M, Nauck M, Wieland H, Perruchoud AP, Roth M. Proliferation of human non-small-cell lung cancer cell lines: role of interleukin-6. Am J Respir Cell Mol Biol. 1998;19(4):606–12.
- Han JH, Lee JH, Han KD, Seo HM, Bang CH, Park YM, et al. Epidemiology and medication trends in patients with psoriasis: a nationwide populationbased cohort study from Korea. Acta Derm Venereol. 2018;98(4):396–400.
- 3. Wang TS, Hsieh CF, Tsai TF. Epidemiology of psoriatic disease and current treatment patterns from 2003 to 2013: a nationwide, population-based observational study in Taiwan. J Dermatol Sci. 2016;84(3):340–5.
- 4. Tsai TF, Ho JC, Song M, Szapary P, Guzzo C, Shen YK, et al. Efficacy and safety of ustekinumab for the treatment of moderate-to-severe psoriasis: a phase III, randomized, placebo-controlled trial in Tai-wanese and Korean patients (PEARL). J Dermatol Sci. 2011;63(3):154–63.
- 5. Chiu HY, Wang TS, Chen PH, Hsu SH, Tsai YC, Tsai TF. Psoriasis in Taiwan: from epidemiology to new treatments. Dermatol Sin. 2018;36:115–23.
- 6. Lee JY, Kang S, Park JS, Jo SJ. Prevalence of psoriasis in Korea: A population-based epidemiological study using the Korean national health insurance database. Ann Dermatol. 2017;29(6):761–7.
- 7. Megna M, Potestio L, Fabbrocini G, Camela E. Treating psoriasis in the elderly: biologics and small molecules. Expert Opin Biol Ther. 2022;22(12): 1503–20.
- 8. Ho SS, Tsai TF. Associations between HLA-Cw1 and systemic treatment response of Asian psoriasis patients. Mol Diagn Ther. 2022;26(5):541–9.
- Kim J, Oh CH, Jeon J, Baek Y, Ahn J, Kim DJ, et al. Molecular phenotyping small (Asian) versus large (Western) plaque psoriasis shows common activation of IL-17 pathway genes but different regulatory gene sets. J Invest Dermatol. 2016;136(1):161–72.

- 10. Kim TG, Lee HJ, Youn JI, Kim TY, Han H. The association of psoriasis with human leukocyte antigens in Korean population and the influence of age of onset and sex. J Invest Dermatol. 2000;114(2):309–13.
- 11. Nakagawa H, Akazaki S, Asahina A, Tokunaga K, Matsuki K, Kuwata S, et al. Study of HLA class I, class II and complement genes (C2, C4A, C4B and BF) in Japanese psoriatics and analysis of a newlyfound high-risk haplotype by pulsed field gel electrophoresis. Arch Dermatol Res. 1991;283(5):281–4.
- 12. Tsai TF, Hu CY, Tsai WL, Chu CY, Lin SJ, Liaw SH, et al. HLA-Cw6 specificity and polymorphic residues are associated with susceptibility among Chinese psoriatics in Taiwan. Arch Dermatol Res. 2002;294(5):214–20.
- 13. Huang YH, Tsai TF. HLA-Cw1 and psoriasis. Am J Clin Dermatol. 2021;22(3):339–47.
- 14. Camela E, Potestio L, Ruggiero A, Ocampo-Garza SS, Fabbrocini G, Megna M. Towards personalized medicine in psoriasis: current progress. Psoriasis (Auckland, NZ). 2022;12:231–50.
- 15. Megna M, Tommasino N, Potestio L, Battista T, Ruggiero A, Noto M, et al. Real-world practice indirect comparison between guselkumab, risankizumab, and tildrakizumab: results from an Italian 28-week retrospective study. J Dermatolog Treat. 2022;33(6):2813–20.
- Sofen H, Smith S, Matheson RT, Leonardi CL, Calderon C, Brodmerkel C, et al. Guselkumab (an IL-23-specific mAb) demonstrates clinical and molecular response in patients with moderate-to-severe psoriasis. J Allergy Clin Immunol. 2014;133(4): 1032–40.
- Nestle FO, Kaplan DH, Barker J. Psoriasis. N Engl J Med. 2009;361(5):496–509.
- Wilson NJ, Boniface K, Chan JR, McKenzie BS, Blumenschein WM, Mattson JD, et al. Development, cytokine profile and function of human interleukin 17-producing helper T cells. Nat Immunol. 2007;8(9):950–7.
- 19. Blauvelt A, Papp KA, Griffiths CE, Randazzo B, Wasfi Y, Shen YK, et al. Efficacy and safety of guselkumab, an anti-interleukin-23 monoclonal antibody, compared with adalimumab for the continuous treatment of patients with moderate to severe psoriasis: Results from the phase III, doubleblinded, placebo- and active comparator-controlled VOYAGE 1 trial. J Am Acad Dermatol. 2017;76(3): 405–17.
- 20. Reich K, Armstrong AW, Foley P, Song M, Wasfi Y, Randazzo B, et al. Efficacy and safety of

guselkumab, an anti-interleukin-23 monoclonal antibody, compared with adalimumab for the treatment of patients with moderate to severe psoriasis with randomized withdrawal and retreatment: Results from the phase III, double-blind, placebo- and active comparator-controlled VOY-AGE 2 trial. J Am Acad Dermatol. 2016;76(3): 418–31.

- 21. Reich K, Griffiths CEM, Gordon KB, Papp KA, Song M, Randazzo B, et al. Maintenance of clinical response and consistent safety profile with up to 3 years of continuous treatment with guselkumab: Results from the VOYAGE 1 and VOYAGE 2 trials. J Am Acad Dermatol. 2020;82(4):936–45.
- 22. Griffiths CEM, Papp KA, Song M, Miller M, You Y, Shen YK, et al. Continuous treatment with guselkumab maintains clinical responses through 4 years in patients with moderate-to-severe psoriasis: results from VOYAGE 1. J Dermatolog Treat. 2022;33(2):848–56.
- 23. Reich K, Gordon KB, Strober BE, Armstrong AW, Miller M, Shen YK, et al. Five-year maintenance of clinical response and improvements in health-related quality of life in patients with moderate-tosevere psoriasis treated with guselkumab: results

from VOYAGE 1 and VOYAGE 2. Br J Dermatol. 2021;185(6):1146–59.

- 24. Reich K, Song M, Li S, Jiang J, Youn SW, Tsai TF, et al. Consistent responses with guselkumab treatment in Asian and non-Asian patients with psoriasis: An analysis from VOYAGE 1 and VOYAGE 2. J Dermatol. 2019;46(12):1141–52.
- 25. Blauvelt A, Tsai TF, Langley RG, Miller M, Shen YK, You Y, et al. Consistent safety profile with up to 5 years of continuous treatment with guselkumab: Pooled analyses from the phase 3 VOYAGE 1 and VOYAGE 2 trials of patients with moderate-to-severe psoriasis. J Am Acad Dermatol. 2022;86(4): 827–34.
- 26. Imafuku S, Zheng M, Tada Y, Zhang X, Theng C, Thevarajah S, et al. Asian consensus on assessment and management of mild to moderate plaque psoriasis with topical therapy. J Dermatol. 2018;45(7): 805–11.
- 27. Megna M, Potestio L, Fabbrocini G, Ruggiero A. Long-term efficacy and safety of guselkumab for moderate to severe psoriasis: a 3-year real-life retrospective study. Psoriasis (Auckland, NZ). 2022;12: 205–12.